Author's response to reviews

**Title:** Efficacy and safety of acupuncture for chronic pain caused by gonarthrosis: A study protocol of an ongoing multi-centre randomised controlled clinical trial [ISRCTN27450856]

**Authors:**

Dr Konrad KS Streitberger (Konrad_Streitberger@med.uni-heidelberg.de)  
Mr Steffen SW Witte (witte@imbi.uni-heidelberg.de)  
Dr Ulrich UM Mansmann (mansmann@imbi.uni-heidelberg.de)  
Christine CK Knauer (knauer@imbi.uni-heidelberg.de)  
Jurgen JK Kramer (juergen.kraemer@ruhr-uni-bochum.de)  
Hanns-Peter HPS Scharf (hanns-peter.scharf@ortho.ma.uni-heidelberg.de)  
Norbert NV Victor (victor@imbi.uni-heidelberg.de)

**Version:** 2  
**Date:** 23 Mar 2004

* Changed Anesthesiology to Anaesthesiology  
* Section headings - Changed from Block capitals to normal  
* Section interventions - "for patients with varus gonarthrosis, cane" deleted  
* Table placement text removed  
* Section changed to "Authors' contributions"  
* Reference 1 - Author changed from Anonymous to [No authors listed].